

Citation: Pigott DM, Howes RE, Wiebe A, Battle KE, Golding N, Gething PW, et al. (2015) Prioritising Infectious Disease Mapping. PLoS Negl Trop Dis 9 (6): e0003756. doi:10.1371/journal.pntd.0003756

Editor: Xiao-Nong Zhou, National Institute of Parasitic Diseases China CDC, CHINA

Received: October 5, 2014

Accepted: April 13, 2015

Published: June 10, 2015

Copyright: © 2015 Pigott et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the paper and its Supporting Information files.

Funding: The initial work to develop a platform that will map a range of infectious diseases is funded by the Bill & Melinda Gates Foundation (Global Health Grant Number OPP1093011), and the authors are grateful to the Surveillance and Vaccine Development team for their contribution to the prioritisation work described in this paper. DMP is funded by a Sir Richard Southwood Graduate Scholarship from the Department of Zoology at the University of Oxford; REH is financially supported by a Wellcome Trust Senior Research Fellowship to SIH (#095066) which also supports AW and KEB; NG is funded by a grant RESEARCH ARTICLE

Prioritising Infectious Disease Mapping

David M. Pigott^{1‡}, Rosalind E. Howes^{1‡}, Antoinette Wiebe¹, Katherine E. Battle¹, Nick Golding², Peter W. Gething¹, Scott F. Dowell³, Tamer H. Farag³, Andres J. Garcia³, Ann M. Kimball³, L. Kendall Krause³, Craig H. Smith³, Simon J. Brooker⁴, Hmwe H. Kyu⁵, Theo Vos⁵, Christopher J. L. Murray⁵, Catherine L. Moyes², Simon I. Hay^{2,5,6}*

 Spatial Ecology & Epidemiology Group, Department of Zoology, University of Oxford, Oxford, United Kingdom, 2 Wellcome Trust Centre for Human Genetics, University of Oxford, Oxford, United Kingdom,
 Bill & Melinda Gates Foundation, Seattle, Washington, United States of America, 4 London School of Hygiene & Tropical Medicine, London, United Kingdom, 5 Institute for Health Metrics and Evaluation, University of Washington, Seattle, Washington, United States of America, 6 Fogarty International Center, National Institutes of Health, Bethesda, Maryland, United States of America

These authors share first authorship on this work.
<u>simon.hay@well.ox.ac.uk</u>

Abstract

Background

Increasing volumes of data and computational capacity afford unprecedented opportunities to scale up infectious disease (ID) mapping for public health uses. Whilst a large number of IDs show global spatial variation, comprehensive knowledge of these geographic patterns is poor. Here we use an objective method to prioritise mapping efforts to begin to address the large deficit in global disease maps currently available.

Methodology/Principal Findings

Automation of ID mapping requires bespoke methodological adjustments tailored to the epidemiological characteristics of different types of diseases. Diseases were therefore grouped into 33 clusters based upon taxonomic divisions and shared epidemiological characteristics. Disability-adjusted life years, derived from the Global Burden of Disease 2013 study, were used as a globally consistent metric of disease burden. A review of global health stakeholders, existing literature and national health priorities was undertaken to assess relative interest in the diseases. The clusters were ranked by combining both metrics, which identified 44 diseases of main concern within 15 principle clusters. Whilst malaria, HIV and tuberculosis were the highest priority due to their considerable burden, the high priority clusters were dominated by neglected tropical diseases and vector-borne parasites.

Conclusions/Significance

A quantitative, easily-updated and flexible framework for prioritising diseases is presented here. The study identifies a possible future strategy for those diseases where significant knowledge gaps remain, as well as recognising those where global mapping programs have already made significant progress. For many conditions, potential shared epidemiological information has yet to be exploited. from the Bill & Melinda Gates Foundation (OPP1053338). SJB is supported by a Wellcome Trust Senior Research Fellowship (#092765) and acknowledges the support of the Bill & Melinda Gates Foundation (OPP1033751). CLM is funded by a grant from the Bill & Melinda Gates Foundation (OPP1093011). The funders had no role in study design, data colleciton and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: I have read the journal's policy and the authors of this manuscript have the following competing interests: SJB is a deputy editor of PLoS Neglected Tropical Diseases.

Author Summary

Maps have long been used to not only visualise, but also to inform infectious disease control efforts, identify and predict areas of greatest risk of specific diseases, and better understand the epidemiology of disease over various spatial scales. In spite of the utilities of such outputs, globally comprehensive maps have been produced for only a handful of infectious diseases. Due to limited resources, it is necessary to define a framework to prioritise which diseases to consider mapping globally. This paper outlines a framework which compares each disease's global burden with its associated interest from the policy community in a data-driven manner which can be used to determine the relative priority of each condition. Malaria, HIV and TB are, unsurprisingly, ranked highest due to their considerable health burden, while the other priority diseases are dominated by neglected tropical diseases and vector-borne diseases. For some conditions, global mapping efforts are already in place, however, for many neglected conditions there still remains a need for high resolution spatial surveys.

Introduction

Maps provide an essential evidence-base to support progress towards global health commitments [1]. For example, they provide important baseline estimates of disease limits [2-7], transmission [8–10] and clinical burden [11–14]; underpin surveillance systems and outbreak tracking [15,16]; help target resource allocation from the macro- [17,18] through the meso-[19–22] to the micro-scale [23]; and inform international travel guidelines [24–26]. Significant developments in mapping techniques have occurred over the last few decades, particularly through the use of species distribution models and model-based geostatistics [1,27]. Similarly, disease data has become more widespread and easier to share [28]. Despite these advances however, a recent review of 355 clinically-significant infectious diseases (IDs) indicated that of the 174 IDs for which an opportunity for mapping was identified, only 4% had been comprehensively mapped [1]. For many of these conditions, there is a significant shortfall between existing maps and what can be achieved with contemporary methods and datasets.

Traditional mapmaking has focussed on a vertical, single-species approach, requiring highly labour intensive, and therefore expensive, manual data identification and assembly [13,21,29–31]. The present era of open-access big data, high computational capacity, and rapid software development offers new opportunities for scaling-up the spatial mapping of IDs, primarily through the automation of data gathering and geopositioning but ultimately also to mapping. The Atlas of Baseline Risk Assessment for Infectious Disease (abbreviated ABRAID, as in, "to awake") is a developing software platform designed to exploit this opportunity and has the ambition to produce continuously updated maps for 174 IDs globally [28]. Realising automation of data retrieval and positioning at this scale is a practically non-trivial but conceptually simple, logistic scaling exercise. In order to automate mapping for each ID so that it is continuously updated and improved as new information becomes available, the spatial inference methods used need to be tailored to each unique ID epidemiology [28]. In some cases this will require disease-specific methodological developments. This requires substantial investment, so an objective and systematic approach is required to determine the order in which IDs are to be mapped.

The first stage in this process is to organise all IDs using a schema based upon shared biological and epidemiological traits; for example, "the mosquito-borne arboviruses". Such groups will likely have similar mapping requirements, enabling synergies in data collation, covariate selection, increased efficiency (*i.e.* in software development), and more robust validation of outputs [32,33]. We refer to these disease groups as "mapping clusters" and they form the basic architecture of the prioritisation process.

To rationally prioritise mapping of these conditions, the diseases within each mapping cluster were evaluated based upon their global burden (both morbidity and mortality), as well as the disease's importance amongst public health stakeholders. Data inputs are quantitative in nature and reliant on either independently derived data or data sourced from entire communities rather than selected expert individuals. Therefore, this proposed framework is unaffected by much of the subjectivity associated with other prioritisation studies, and also provides a platform for rapidly incorporating changes to existing diseases, as well as emerging novel public health threats. The prioritisation exercise helps to guide the order in which diseases are mapped to best support public health priorities; we argue that all relevant diseases can and should eventually be mapped.

A comprehensive atlas of IDs is of central importance in providing geographical context to the understanding of tropical disease and global health [34-36]. Moreover, as the atlas becomes more complete the overlay of maps will provide opportunities for investigating patterns of global disease diversity [37,38] and the process of disease emergence [39].

Methods

Method summary

In order to generate disease prioritisation standards, diseases with shared taxonomy and transmission characteristics were grouped together to create clusters. Diseases within each cluster were evaluated based upon two factors reflecting their importance from a public health perspective: (a) the global burden of the disease and (b) the current public health focus on the condition. Both metrics were assessed simultaneously in order to rank the clusters, and specific diseases were then identified for prioritisation.

Selection of diseases for mapping

This study aimed to be comprehensive in its scope of IDs. All diseases identified in a previous review as meriting mapping were included [1]. This earlier study categorised 355 diseases into five classes: Option 1, indicating that the disease was unsuitable for occurrence based mapping methods; Option 2, mapping the observed occurrence of the disease; Option 3, mapping the maximum potential range of the disease using knowledge of vector, intermediate host and reservoir species; Option 4, using niche mapping methods such as boosted regression trees; and Option 5, where sufficient data exist to allow for global maps of variation in prevalence of infection and/or disease. Option 1 diseases included those that showed no sustained spatial variation in occurrence (*i.e.* had a cosmopolitan distribution) and had insufficient evidence to allow for the global mapping of variation in prevalence using advanced statistical methods such as model-based geostatistics. In cases where such information does exist, these diseases were promoted to Option 5 status. Revisions to the *Hay et al. (2013)* paper have led to the inclusion of tuberculosis, ascariasis, trichuriasis and trachoma-all previously listed as Option 1-as Option 5 diseases. Further revisions included the exclusion of New and Old World Spotted Fever Rickettsiosis and New and Old World Phlebovirus because their constituent diseases were included. In addition, Plasmodium knowlesi was included due to the increasing appreciation of its significance to human health in Southeast Asia [40,41]. The new revised total of diseases

that warrant mapping was therefore 176. Those diseases not considered for mapping due to Option 1 classification are outlined and justified in the Supporting Information.

Creating disease clusters

Diseases were grouped into clusters based on characteristics relevant to spatial epidemiology. Diseases were placed in the same cluster if they had the potential to mutually reinforce each other in terms of data assembly, mapping requirements and cross-validation of data by comparison of outputs. Clustering classifications were therefore based on the key factors influencing the approach taken for mapping.

At the coarsest level, pathogens were grouped by agent type (virus/bacteria/fungus/other) and the larger agent groupings were split into specific phyla (*e.g.* Nematoda and Platyhelminths) [42]. These relatively coarse groups reflect fundamental differences in life histories and epidemiology as well as the most basic taxonomic divisions. Within these broad groupings, the mode of transmission was used to create the final disease clusters. This is an important factor when mapping IDs, as the mode of transmission has a large influence on which abiotic correlates are relevant to the mapping process. For instance, the transmission limits of vectorborne diseases are restricted in part by the environmental suitability for the vector species in question, thus diseases spread by similar vectors will share covariates [43]. Similarly, sexuallytransmitted diseases are likely to share mapping methods linked to human distribution and behaviour, whilst pathogens spread by water contact would share common traits linked to the environment; these groupings can therefore be logically considered together within a mapping framework. The mode of transmission classifications are defined in the previous publication [1].

Assessing disease burden

The burden of each disease was assessed using the disability-adjusted life year (DALY) estimates from the 2013 Global Burden of Disease Study (GBD 2013) [44,45]. DALYs quantify both morbidity and mortality attributed to each disease and therefore better capture the total impact of a disease than do clinical cases or mortality alone [44,45]. The GBD's systematic approach across a wide spectrum of diseases provides an extremely valuable resource from which to compare the relative impact of diseases on human health.

Wherever possible, direct links were made between the GBD estimates and diseases in the mapping list. The GBD disease categories, which are based upon the International Classification of Diseases and Related Health Problems (ICD-10) [46], do not always specify particular infectious agents, but rather focus on the clinical symptoms of infection, or non-specified disease groups. These aggregated DALY estimates had to be split across the relevant causative diseases in the mapping list, therefore the *Hay et al. (2013)* study was reconciled with the ICD-10 codes and then GBD categories in order to disaggregate the broader classifications such as "other diarrheal diseases" and "other neglected tropical diseases". The full process is outlined in the associated Supporting Information, <u>S1 Text</u>.

Overall, 11 of the 176 mapping diseases could not be reconciled to the GBD categories. Some were not considered due to having unknown pathogenic agents (*e.g.* tropical sprue) and others were very rare and fell into ICD-10 categories that were assigned over various groupings (*e.g.* pentastomiasis). These diseases were allocated a nominal DALY of 100; this value, while arbitrary, is low enough to avoid skewing the analysis. For each cluster, the total DALYs for all diseases was calculated and contributed to the final analysis.

Assessing global health community interest

Of equal importance is the need to produce maps for those diseases where there is the greatest demand, whether from international organisations or from local public health authorities. Measuring this factor was achieved by surveying a representative subset of potential end-users, to identify which diseases have been prioritised by major public health stakeholders: state-funded public health agencies, private companies (*e.g.* vaccine developers), political bodies, non-governmental advocates and practitioners, as well as the scientific research community. For each disease, the final policy score was the sum of three component scores: public health, stakeholder interest, status as a notifiable disease, and h-index.

Cases from the different categories of public health stakeholders were included to capture the spectrum of interest groups (see <u>S1 Text</u> for full listing). Each organisation's mission statement and project pages were reviewed to identify the diseases contained in their public health portfolio. Depending on the type of stakeholder, this would indicate that the organisation would, for example, dedicate funding and effort towards control of that disease, advocate for the disease to governments or public health agencies, or dedicate research funding to the disease. Each disease was allocated one point per stakeholder reporting an interest in it. An inclusive approach was followed, whereby diseases were considered to be of interest to a stakeholder, irrespective of any hierarchy within the agency's prioritisation system.

Another point was allocated to diseases which were notifiable to national disease control agencies. In order to mitigate spatial bias in the notifiable disease listed by different agencies, a search for countries which had readily-accessible and clearly defined domestic policy relating to named pathogens was performed, and one country from each of the main GBD defined regions was selected: USA (High Income), Brazil (Latin America and the Caribbean), Zambia (sub-Saharan Africa), United Arab Emirates (North Africa and the Middle East), India (South Asia), Malaysia (South East Asia, East Asia and Oceania) and Croatia (Central and Eastern Europe and Central Asia). Interest in these diseases at a domestic level suggests that there will be interest in maps of these diseases, as demonstrated by the presence of subscription-only online databases of maps including GIDEON [47] and the rapid expansion of real-time maps to which physicians are encouraged to contribute [15,28].

Academic output, a proxy of funder agency awards, but also of high-quality data availability [1], was quantified based on the *h*-index of each disease [48], as reported by Scopus [49]. More commonly used to assess a scientist's productivity and impact, the *h*-index is used here to quantify the level of active interest across the academic community in each disease [50]. The *h*-index is the number of published papers (referring to a particular disease) that have been cited by at least as many other papers. In other words, an *h*-index of 7 signifies that 7 published papers including that disease name have been cited at least 7 times. For each disease (document search for "Disease Name" in "Article Title, Abstract, Keywords"). This Scopus search generates a Citation Tracker file showing the number of citations to each publication referring to the "Disease Name". Diseases were then categorised according to their *h*-index. Those for which there was evidence of very high scientific output scored 2 (*h*-index >100), those with intermediate *h*-index (>50–100) scored 1.5, while diseases with *h*-index of <50 scored 1.

The diseases classified as Option 4 (use niche modelling methods) and Option 5 (model prevalence or incidence) have the most epidemiological data available and have the greatest potential to benefit from a dedicated mapping exercise, but also require the most resources. Option 2 and 3 diseases are data-poor and both require mapping of occurrence data only [1], and therefore are significantly less time-intensive to map, limited to more simplistic analyses, than those diseases categorised as Option 4 and 5. Option 3 disease mapping relates potential

transmission limits to aspects of vector biology. In cases where Option 4 and 5 diseases also have the same vector, the Option 3 disease will be considered as part of mapping these complementary diseases; where this is not the case, a disease's transmission limits can be assessed through a mixture of literature surveys and occurrence data overlap. Option 4 and 5 diseases within the disease clusters were therefore prioritised and for each cluster, the average policy score for the Option 4 and 5 diseases was calculated and contributed to the final analysis. These diseases should be the primary focus of future cartographic efforts as these require the most attention and bespoke inputs to be generated.

Mapping prioritisation ranking of diseases

The final step in the process was to combine these assessments to produce a ranking of disease clusters and therefore recommend diseases to prioritise for mapping. Each cluster was plotted on a graph based on its total DALYs and the average policy priority of its Option 4 and 5 diseases. Option 2 and 3 diseases were included in the cluster DALY scores in order to reflect the relative importance that each cluster represented in terms of burden of disease. One cluster may consist of a large number of minor diseases which, as a collective grouping, represent a significant problem—by retaining the DALY score, this burden is reflected, With the policy priority score however, the opposite is the case; inclusion of multiple low scoring diseases would down-weight the cluster as a whole. In scenarios where clusters consist of a diverse grouping of pathogens, averaging policy score across all conditions misrepresents those with a high policy priority and therefore masks these diseases in comparison to clusters that only consist of those diseases with high policy priority scores.

Each cluster was then evaluated based upon its distance from a hypothetical cluster which had the highest DALYs (*i.e.* that of HIV) and the highest policy score (*i.e.* that of Malaria) relative to a line drawn from this cluster to the origin; those closer to this hypothetical cluster, along this axis, were prioritised higher. As a result, the relative influence of burden and policy priority could be considered both simultaneously and independently. Within each cluster, the diseases to be prioritised (*i.e.* Option 4 or 5) were then reported (<u>Table 1</u>). The code to replicate this methodology is freely available from: <u>https://github.com/SEEG-Oxford/prioritisation</u>.

Results

Organization of the mapping clusters

The 176 diseases identified as having a rationale for mapping were organised into 33 clusters, based upon the biological and taxonomic classifications of the causative pathogen, modes of transmission and the mapping method recommended in a previous review [1] (Fig 1). Seven of these clusters included only a single disease due to their unique transmission within their broader taxonomic grouping (HIV, poliomyelitis, avian influenza, pythiosis, South American bartonellosis, tuberculosis and babesiosis). Conversely, the mosquito-borne arbovirus cluster was the largest cluster, consisting of 26 diseases, many of which have the potential to benefit from modelled maps.

Prioritising mapping diseases

Fig 2 brings together the two indices selected to prioritise diseases for mapping—disability adjusted life-year (DALY) burden and relative stakeholder interest. These plots demonstrate that the HIV, malaria and tuberculosis clusters are exceptional in representing an overwhelming share of DALY burden [51] and being of highest priority to the global health community with their placement in the top right quadrant of the graph. These three clusters contain five Table 1. Clusters indicated as mapping priorities with their constituent diseases recommended for distribution modelling and current global mapping projects identified.

Cluster (main diseases to map / total diseases in cluster)	Diseases within cluster, to map	Total cluster DALYs	Average policy score	Current global mapping projects
1. Malaria (n = 3/5)	Plasmodium falciparum	65,493,135	11.8	MAP [<u>13,29,40];</u> WHO [<u>81]</u>
	P. knowlesi			
	P. vivax			
2. HIV (n = 1/1)	HIV	69,480,661	11	GBD [<u>51</u>]; UNAIDS [<u>82]</u>
3. Tuberculosis (n = 1/1)	Tuberculosis	49,816,215	11	GBD [<u>51]</u>
4. Food/Water-borne (Bacteria) (n = 1/4)	Cholera	9,962,003	8	
5. Water-borne (Platyhelminth) (n = $3/7$)	Schistosoma haematobium	3,062,843	7.7	GAHI [<u>83];</u> Global NTD database [<u>31]</u>
	S. japonicum			
	S. mansoni			
6. Trypanosomiasis (n = 2/2)	African trypanosomiasis	728,564	7.5	WHO [<u>68,84]</u>
	American trypanosomiasis			
7. Filariasis (n = $3/3$)	Bancroftian filariasis	2,022,099	6.2	GAHI [<u>85</u>]
	Brugia malayi B. timori			
8. Soil Transmitted Helminths (n = $3/3$)	Ascariasis	4.029.403	5.3	GAHI [5.14]
	Hookworm	.,020,100	0.0	
	Trichuriasis			
9. Leishmaniasis (n = 3/3)	Cutaneous leishmaniasis (Old World)	4,283,139	5.2	SEEG [8]
	Cutaneous leishmaniasis (New World)			
	Visceral leishmaniasis			
10. Unknown agent (n = 1/4)	Tropical sprue	3,609,400	4	
11. Picornaviridae (n = 1/1)	Polio	116,065	6	The Global Polio Eradication Initiative [<u>86]</u>
12. Food/Water-borne (Nematode) ($n = 1/13$)	Dracunculiasis	422,476	2.5	
13. Fly-borne (Nematode) (n = $2/5$)	Loiasis	711,246	4.3	WHO and APOC [87] [88]
	Onchocerciasis			
14. Direct contact (Bacteria) (n = 4/6)	Anthrax	1,030,777	4	
	Brazilian purpuric fever			
	Leprosy			
	Trachoma			GAT [<u>10,89]</u>
15. Mosquito-borne (Virus) (n = 15/26)	Barmah Forest disease	4,219,569	2.6	
	California serogroup viruses			
	Chikungunya			
	Dengue			SEEG [2,12]
	Japanese encephalitis			
	Murray Valley encephalitis			
	Rift Valley fever			
	Rocio			
	Ross River virus			
	Sindbis			
	St. Louis encephalitis			
	Venezuelan equine encephalitis			
	Western equine encephalitis			

(Continued)

Table 1. (Continued)

Cluster (main diseases to map / total diseases in cluster)	Diseases within cluster, to map	Total cluster DALYs	Average policy score	Current global mapping projects
	West Nile fever			
	Yellow fever			

* Indicates default null value.

MAP—Malaria Atlas Project; WHO—World Health Organization; GBD—Global Burden of Disease; GAHI—Global Atlas of Helminth Infections; SEEG— Spatial Ecology and Epidemiology Group; APOC—African Programme for Onchocerciasis Control; GAT—Global Atlas of Trachoma

doi:10.1371/journal.pntd.0003756.t001

individual diseases that are a mapping priority, malaria (*Plasmodium falciparum*, *P. vivax*, and *P. knowlesi*), HIV and tuberculosis. <u>Table 1</u> shows the top 15 disease clusters (*i.e.* those in the top right of Fig 2), representing 44 individual diseases, with their associated scores. Fig 3 demonstrates that there exists a group of approximately 45 diseases that are the collective focus of public health agencies. The 44 diseases prioritised by this study include all those diseases that represent a significant cartographic challenge (*i.e.* those diseases requiring either species distribution modelling approaches to produce occurrence maps or model-based geostatistics to produce prevalence maps, n = 33) identified by these public health agencies, save rabies and avian influenza. The clusters are ranked in order, whilst the diseases within each cluster are alphabetical and should be considered equal on the basis of this prioritisation.

The top ten priority clusters account for over 92% of all DALYs for those IDs which require mapping (*i.e.* the 176 IDs identified); if this is expanded to the top 15 clusters containing 44 diseases to map, this value increases to 95% (Fig 4). Within these 44 diseases, 19 of the 29 neglected tropical diseases (NTD) highlighted by the WHO are represented. Within the top ten prioritised clusters, 14 individual diseases relate to these same NTDs [52,53]. The top 15 prioritised clusters include some diseases, such as the picornaviridae (polio), that have a low DALY burden but a high public health ranking because they are high on the eradication agenda.

Disease burden

It was possible to establish a direct correspondence with GBD estimates for 34 of the 176 diseases with a strong rationale for mapping as listed by *Hay et al. (2013)* [44,45]. DALY estimates were allocated to a further 132 diseases by linking diseases with ICD-10 codes [46] and their respective GBD category definitions. Whilst these burden values are not accurate absolute values, and should not be interpreted as such, this DALY allocation does allow relative burdens to be determined. The remaining 11 diseases were given the baseline DALY allocation of 100, a value not intended to represent an estimate of the "true" DALYs associated with these diseases, but rather to distinguish them from diseases which were considered to cause a major burden in the GBD analysis. It is safe to assume that if such diseases were not assigned a specific GBD classification, their global impact on mortality and morbidity is relatively small.

In total, the 176 diseases with a strong rationale for mapping [1] represent over 230 million DALYs, approximately 10% of the global DALY burden and 47% of the global ID DALY burden. At the cluster level, HIV, malaria and tuberculosis represent 80% of the overall mapping-disease DALY burden (Fig 5A). Apart from these three conditions, the only other IDs in the top 50 highest DALYs globally are not currently recommended for mapping because they do not show spatial variation in their occurrence and have insufficient data to map variation in disease prevalence with model-based geostatistical analyses. The high-burden diseases not currently considered for mapping include respiratory diseases, meningitis, and many diarrhoeal



Fig 1. Hierarchical organisation of the 33 clusters. The 176 diseases with strong rationale for mapping were first sorted by taxonomy of pathogenic agent (in orange) and then structured by common epidemiological and transmission characteristics into sub-groupings (in blue) and finally clusters (in red). STH = soil transmitted helminth, VBD = vector borne disease.

doi:10.1371/journal.pntd.0003756.g001

infections. Alternative approaches to mapping broader symptom groupings (severe pneumonia, severe diarrhoea and severe febrile illnesses) and then differentiating constituent disease



Fig 2. Disease prioritisation. Plot showing the 33 clusters of diseases as ranked by burden of disease DALYs (y-axis—logarithmic scale) and mean policy priority score of occurrence mapping and prevalence mapping diseases (x-axis—linear scale). The top ten clusters circled and numbered as identified in <u>Table 1</u>. The size of the circle is determined by the total number of diseases contained and colour is based upon taxonomy (as outlined by <u>Fig 1</u>; the web appendix contains the full disease listing for each cluster). The dashed guidelines are perpendicular to the axis along which prioritisation order for the clusters was determined; those closer to the top right, along this axis, were prioritised higher.

doi:10.1371/journal.pntd.0003756.g002



Fig 3. A "species accumulation" curve showing the cumulative number of diseases of interest sampled by increasing numbers of public health stakeholders examined. The diseases of interest of twenty global health stakeholders was indexed and plotted (see <u>Methods</u>). As additional organisations are sampled beyond the fifteen used in this study, the number of unique diseases identified plateaus at around 42. Thus not all public health stakeholders need to be sampled to capture the global diversity of diseases of public health interest.

doi:10.1371/journal.pntd.0003756.g003

components, are being developed. Together, this would map 80% of all DALYs caused by communicable diseases.

A higher resolution focus on the clusters excluding HIV, malaria and tuberculosis (Fig 5B) shows that over 60% of DALYs associated with the 176 IDs are accounted for by the other top ten prioritised clusters; approximately three quarters of the remaining DALYs are accounted for when the remaining prioritised clusters of diseases are included.

Global health community interest

The treemap in Fig 5C displays the repartition of interest from the global health community across the clusters. Interest was scored in terms of: 1) the stated priorities of a survey of assorted public health stakeholders who are expected to be end-users of the maps, 2) status as a notifiable disease, and 3) prominence in the academic literature.

A total of 20 diverse stakeholders were surveyed. This was found to be a sufficiently large number to sample based on an analysis similar to a species accumulation curve that demonstrates the diminishing returns from increasing sampling effort [54]. The number of new



Fig 4. Cumulative percentage barplot indicating the cumulative percentage of DALYs accounted for by each cluster. The colouring is based upon taxonomy, as in Fig 2. The red line indicates the top ten clusters, the dark green indicates the top 15.

doi:10.1371/journal.pntd.0003756.g004

diseases reported levelled off at around 15 organisations sampled (Fig 3) and so the 20 organisations used for this analysis was sufficient to capture the diseases of public health priority. Of the 176 diseases recommended for mapping [1], 24% were prioritised by at least one public health agency, and 55% were notifiable to at least one of the national disease control agencies.

Of those diseases that represent the greatest cartographic challenge, all were prioritised by at least one public health agency and two thirds were notifiable diseases. Of the 176 diseases, thirty diseases (17%) had an *h*-index [48] above 100 (with HIV having the highest *h*-index of 461), while 64% of the diseases had an *h*-index of 50 or less. Of the occurrence mapping and prevalence mapping diseases, 30% had an *h*-index above 100 and only 37% had an *h*-index of 50 or less.

Unlike the DALY burden, which was allocated at the disease level (<u>S1 Text</u>), the stated priority diseases were often grouped to the cluster level by the surveyed stakeholders. For instance, rather than specifying "*Plasmodium vivax*" or "visceral leishmaniasis" as a focus, "malaria" and "leishmaniasis" would be more commonly stated targets. Each component disease of these clusters would therefore be allocated a point, meaning that the number of component diseases in each cluster strongly inflated the overall interest score allocated at the cluster aggregate. Interest scores were calibrated in the final prioritisation assessment to the number of diseases classified as occurrence or prevalence mapping within each cluster (*i.e.* those requiring the more advanced geostatistical techniques, see <u>Methods</u> for more details), so as to avoid being unduly skewed by the size of the cluster.

Overall, malaria, HIV and tuberculosis were the leading clusters of interest, with scores of 11.8, 11 and 11, respectively. A further seven clusters received repeated interest, including



Fig 5. Plots indicating the relative importance of each mapping cluster. (A) Area of each section is determined by the total DALY contribution of each of the 33 clusters. Blue indicates a cluster contributing to the top ten clusters to be prioritised, green indicates top 44 diseases (n = 5 clusters) and light green represents the remaining disease clusters (n = 18). (B) Area of each section is determined by the total DALY contribution of 30 clusters, with HIV, tuberculosis and malaria excluded. Blue indicates a cluster contributing to the top ten clusters to be prioritised (n = 7), green indicates top 44 diseases (n = 5 clusters) and light green represents the remaining disease clusters (n = 18). (B) Area of each section is determined by the total DALY contribution of 30 clusters, with HIV, tuberculosis and malaria excluded. Blue indicates a cluster contributing to the top ten clusters to be prioritised (n = 7), green indicates top 44 diseases (n = 5 clusters) and light green represents the remaining disease clusters (n = 18). STH = soil-transmitted helminth, (B)—bacteria, (N)— nematode, (PI)—platyhelminth, (V)—virus. (C) Area of each section is determined by the total policy interest score of each of the 33 clusters. Red indicates a cluster within the top ten to be prioritised, orange indicates one of top 44 diseases (n = 5) and light pink represents the remaining disease clusters (n = 18). STH = soil-transmitted helminth, (B)—bacteria, (N)—nematode, (PI)—platyhelminth, (V)—virus.

doi:10.1371/journal.pntd.0003756.g005

food-borne/water-borne bacteria (score = 8) and water-borne trematodes (7.7), trypanosomiasis (7.5), filariasis (6.2), picornaviridae (6), avian contact viruses (6), soil-transmitted helminths (5.3) and leishmaniasis (5.2) all scoring highly, indicating their importance to the public health community. These scores are relative and intended to reveal general trends across the clusters rather than quantitatively reflect the weighting that any one institution places on a particular disease.

Discussion

A review of all clinically significant IDs identified 176 with a strong rationale for mapping, of which only 4% have been adequately mapped [1]. The current study was undertaken to define a ruleset for determining which diseases, from a cartographic and public health perspective, should be prioritised when sequentially addressing this shortfall. Diseases were clustered together based upon shared characteristics (such as basic taxonomic division and mode of transmission) in order to consider together those diseases that would synergise operationally in terms of data collection, covariate selection and methodology used. Given the large number of diseases identified, prioritisation is necessary; we addressed this by evaluating both within the context of disease burden as well as considering the diseases' influence within public health organisations and the wider academic community. It is important to stress that the study was focussed on priorities for mapping, and was not a general prioritisation of IDs; this is particularly important to emphasise given that a number of high-burden diseases, including meningitis, pneumonia and some diarrhoeal diseases, were not included in the list of 176 diseases [1,44,45].

Malaria is the infectious disease for which the most detailed and robust global risk maps exist [13,29]. The work of the Malaria Atlas Project [33,55] along with a proliferation of national and local-scale studies [56] has established a mature and sophisticated methodological approach centred on the use of model-based geostatistics to generate continuous surfaces of risk. This has been possible, in part, due to the long history of population-based malaria infection prevalence surveys where researchers and control programmes have used microscopy or rapid diagnostic tests to establish the proportion of randomly sampled individuals testing positive for malaria parasitaemia [30,57]. Crucial for geospatial mapping, such data are increasingly georeferenced with a latitude and longitude for each observation established *via* gazetteer methods (recorded location names linked to digital atlases) or directly using Global Positioning System (GPS) technology at the time of survey [58,59].

The high prioritisation of HIV and tuberculosis shown in the current study brings into sharp focus the need for similar mapping activities to be established for HIV and tuberculosis. All three diseases have an established history of routine and survey-based data collection that, in comparison to many other diseases, is of relatively high quality and consistency, laying the foundation for similar statistical mapping approaches to those used for malaria to be applied. A cornerstone of HIV surveillance over the last several decades has been routine blood testing for HIV infection in mothers attending sentinel antenatal clinics. Such data provide rich longitudinal observations of prevalence in this demographic group and the potential exists to combine these with cross-sectional data from nationally representative household surveys [60] to generate optimal space-time models of the changing geographical pattern of infection across individual countries. Unlike HIV and malaria, population-based tuberculosis prevalence testing is not currently included as part of the major international survey programmes [58,61]. However, such surveys (reporting on the prevalence of bacteriologically-confirmed pulmonary tuberculosis) have been undertaken in a number of high-burden countries in recent years, with many more planned in the near future [62]. In a similar way to HIV, the prospect exists of a mapping methodology that could combine survey-based data with the rich health-system based data on new case notifications and other metrics, leveraging the respective strengths of community- and facility-based data. A longer-term goal must be the development of a data assimilation and modelling architecture for all three of these major global diseases to support robust and regularly updated global maps detailing their joint distribution and its evolution though time which can be used to assess the impact of control and international financing efforts [18].

The current analysis identifies a number of different NTDs as priority diseases for mapping, a finding which is consistent with the emphasis given to mapping by the global NTD community in order to geographically target NTDs interventions [63,64]. Specifically, for those NTDs where morbidity control is the goal, including soil-transmitted helminths (STH) and schistosomiasis, interventions are most cost-effective when they are targeted to areas of highest transmission [21]. For those NTDs which are identified for elimination, such as onchocerciasis and lymphatic filariasis, it is essential to know where transmission occurs and when it has been successfully halted following control measures. As a consequence of these operational requirements, large-scale mapping initiatives are underway for each of the main NTDs (Table 1). A challenge for mapping the NTDs, and indeed for mapping many IDs, is the need to continually update maps in order to help track the progress in control. As interventions reduce transmission levels and therefore distributions become more focalised, the need for mapping will only increase.

Unsurprisingly, the top 44 diseases for prioritisation are dominated by those with the highest global burden. However, certain clusters stand out as having high public health attention without a high burden, particularly the picornaviridae cluster and its constituent disease, polio. Although cases are now restricted to a few hundred each year, polio has been identified as an eradication target and is a high priority for many public health stakeholders despite recent obstacles in the eradication schedule [65,66]. In these eradication and elimination scenarios, the role of mapping changes subtly to both identifying areas where cases continue to occur, and in highlighting potential future risks and improving surveillance [67]. Following a similar logic, diseases such as dracunculiasis, African trypanosomiasis and onchocerciasis, in spite of relatively low burdens, remain high policy priorities due to elimination efforts in various parts of the globe [68,69]. These examples demonstrate the utility of the approach used in this study of using assessments of the public health burden as well as metrics of public health attention.

The disease prioritisation methodology used here differs from existing approaches, such as the "Delphi panel method", in that it does not include a panel of experts scoring various criteria associated with the diseases being considered [70-74]. In contrast, this study uses a simplified methodology, placing importance in reproducibility and flexibility, using clearly defined rules to assess available evidence and remove potentially subjective expert-opinion. The methods employed are reliant on independent, third party information, and are assessed in a consistent manner, which can easily respond to changes either in burden or public health focus. The

relative importance of these diseases will most likely change over time, so an approach that can easily accommodate this is preferable. Burden estimation using the GBD is crucial, since it is the leading globally consistent measure by which to compare these various diseases and the effects of their many different clinical manifestations. Any global assessment of 301 causes of mortality and morbidity, and associated sequelae, will be subject to the limitations of data availability and epidemiological understanding as well as model assumptions and implementation [53,75,76], and will require frequent updates in a rapidly changing world. The technique presented here has the advantage of being rapidly updateable, and we will reproduce these numbers with each new iteration of the GBD project. As a consequence, public health authorities can also easily create bespoke prioritisation lists based upon a selection of disease inclusion criteria (such as those endemic to their particular country or region). This can more easily be achieved with the availability of sub-national estimates of disease burden from the GBD study. Country specific estimates of the interest scores can also be generated with greater specificity, and can therefore avoid some of the potential biases resulting from the use of other countries as representatives of each GBD region used in this study.

Additional factors that may influence the disease priority, such as potential economic impact [77–79], were not used in this analysis because insufficient information was available to include these metrics. The methodology outlined above benefits from two metrics that can be applied globally to quantify DALYs and public health priority. As and when measures of additional disease impacts become available, they can and should be incorporated into assessments such as this.

The study also identifies some high DALY groupings that do not have high-level policy interest. Three groupings (Tick-borne (Bacterial), Tick-borne (Viral) and Mammal contact (Viral)) have a cumulative high DALY burden, but relatively low policy rankings and therefore are just outside the top 15 cluster listing. This may reflect the large number of diverse pathogens that make up these groupings, many of which are relatively restricted in distribution and hence would not commonly be prioritised by globally focussed organisations. That said, the high DALY value indicates that these diseases are of international interest, particularly when secondary human-to-human transmission is a possibility such as with Lassa fever and Crimean Congo Haemorrhagic Fever [80]. These conditions further advocate the utility of regional and national level priority estimates.

The exclusion of diseases not suited for occurrence based mapping, and therefore omitted from the prioritisation process (so called Option 1 diseases [1]), is entirely based on cartographic considerations. Some of these diseases are inherently linked to human-to-human interactions, others are endogenous in origin, with the pathogen essentially ubiquitous amongst humans and only occasionally causing opportunist infections in certain scenarios, whilst some have the potential to cause infection anywhere across the globe due to the cosmopolitan distribution of their sources of infection, whether they be environmental or human based. Many of these diseases can vary spatially, as evidenced by the African meningitis belt, although such variation, when considered relative to the rest of the world, is due to differences in prevalence or intensity, not presence or absence. Occurrence based mapping methods, such as boosted regression trees, rely on binary presence/absence data. For conditions such as the common cold, diphtheria or respiratory syncytial virus, which have the potential to occur across the globe, these mapping techniques are ineffective. It is only through using more advanced methods, such as model-based geostatistics, that maps analysing the variation in intensity of these diseases can be produced. The limitation of this methodology is the amount of prevalence survey data required, which for many diseases is not comprehensive or detailed enough to allow for global analyses. Basic human related covariates, such as population density, urban extent profiles and national vaccination statistics can be used to explain a degree of the global variation in

these diseases, but fall short of the wealth of information that can be derived from comprehensive global prevalence datasets, such as those available for malaria. As we continue to explore additional data avenues, there will be an increasing number of diseases where such data become available.

The disease prioritisation outlined in this study offers a logical framework for proceeding with disease mapping, which reinforces the necessity of existing programmes and identifies those diseases to focus on next (Table 1). Diseases which will form the initial focus of future study comprise both those with the highest-burden and those of greatest concern to the global health community. The initial top-priority diseases include a range of disease agents and transmission routes, and therefore present a variety of challenges for mapping. The prioritisation and clustering of these diseases presents a clear plan of action designed to maximise the effectiveness and value of future cartographic efforts.

Supporting Information

S1 Text. Supporting Information providing more specific details on the rationale for mapping, linking the Global Burden of Disease and those identified as mapping targets, feeds for the identifying diseases of interest to public health stakeholders and a full cluster ranking. (DOCX)

Acknowledgments

We thank Maria Devine, Kirsten Duda and Moritz Kraemer for proofreading.

Author Contributions

Conceived and designed the experiments: SIH. Performed the experiments: REH DMP AW. Analyzed the data: REH DMP KEB AW NG. Contributed reagents/materials/analysis tools: SJB AW DMP REH HHK TV CJLM. Wrote the paper: DMP REH AW KEB NG PWG THF AJG AMK CHS SJB CLM SIH SFD LKK HHK TV CJLM.

References

- 1. Hay SI, Battle KE, Pigott DM, Smith DL, Moyes CL, et al. (2013) Global mapping of infectious disease. Philos Trans R Soc Lond B Biol Sci 368: 20120250. doi: 10.1098/rstb.2012.0250 PMID: 23382431
- Brady OJ, Gething PW, Bhatt S, Messina JP, Brownstein JS, et al. (2012) Refining the global spatial limits of dengue virus transmission by evidence-based consensus. PLoS Negl Trop Dis 6: e1760. doi: <u>10.1371/journal.pntd.0001760</u> PMID: <u>22880140</u>
- Guerra CA, Gikandi PW, Tatem AJ, Noor AM, Smith DL, et al. (2008) The limits and intensity of *Plasmo-dium falciparum* transmission: Implications for malaria control and elimination worldwide. PLoS Med 5: 300–311.
- Guerra CA, Howes RE, Patil AP, Gething PW, Van Boeckel TP, et al. (2010) The international limits and population at risk of *Plasmodium vivax* transmission in 2009. PLoS Negl Trop Dis 4: e774. doi: <u>10.</u> <u>1371/journal.pntd.0000774</u> PMID: <u>20689816</u>
- Pullan RL, Brooker SJ (2012) The global limits and population at risk of soil-transmitted helminth infections in 2010. Parasit Vectors 5: 81. doi: <u>10.1186/1756-3305-5-81</u> PMID: <u>22537799</u>
- 6. Ellis CK, Carroll DS, Lash RR, Peterson AT, Damon IK, et al. (2012) Ecology and geography of human monkeypox case occurrences across Africa. J Wildl Dis 48: 335–347. PMID: <u>22493109</u>
- 7. Cano J, Rebollo MP, Golding N, Pullan RL, Crellen T, et al. (2014) The global distribution and transmission limits of lymphatic filariasis: past and present. Parasit Vectors 7: 466. PMID: <u>25303991</u>
- 8. Pigott DM, Bhatt S, Golding N, Duda KA, Battle KE, et al. (2014) Global distribution maps of the leishmaniases. eLife: e02851.

- 9. Fichet-Calvet E, Rogers DJ (2009) Risk maps of Lassa fever in West Africa. PLoS Negl Trop Dis 3: e388. doi: 10.1371/journal.pntd.0000388 PMID: 19255625
- Smith JL, Flueckiger RM, Hooper PJ, Polack S, Cromwell EA, et al. (2013) The geographical distribution and burden of trachoma in Africa. PLoS Negl Trop Dis 7: e2359. doi: <u>10.1371/journal.pntd.</u> <u>0002359</u> PMID: <u>23951378</u>
- Hay SI, Okiro EA, Gething PW, Patil AP, Tatem AJ, et al. (2010) Estimating the global clinical burden of *Plasmodium falciparum* malaria in 2007. PLoS Med 7: e1000290. doi: <u>10.1371/journal.pmed.1000290</u> PMID: <u>20563310</u>
- Bhatt S, Gething PW, Brady OJ, Messina JP, Farlow AW, et al. (2013) The global distribution and burden of dengue. Nature 496: 504–507. doi: <u>10.1038/nature12060</u> PMID: <u>23563266</u>
- Gething PW, Elyazar IRF, Moyes CL, Smith DL, Battle KE, et al. (2012) A long neglected world malaria map: *Plasmodium vivax* endemicity in 2010. PLoS Negl Trop Dis 6: e1814. doi: <u>10.1371/journal.pntd.</u> 0001814 PMID: 22970336
- Pullan RL, Smith JL, Jasrasaria R, Brooker SJ (2014) Global numbers of infection and disease burden of soil transmitted helminth infections in 2010. Parasit Vectors 7: 37. doi: <u>10.1186/1756-3305-7-37</u> PMID: 24447578
- Brownstein JS, Freifeld CC, Madoff LC (2009) Digital disease detection—harnessing the web for public health surveillance. N Engl J Med 360: 2153–2157. doi: <u>10.1056/NEJMp0900702</u> PMID: <u>19423867</u>
- Freifeld CC, Mandl KD, Ras BY, Bronwnstein JS (2008) HealthMap: global infectious disease monitoring through automated classification and visualization of internet media reports. J Am Med Inform Assn 15: 150–157. PMID: <u>18096908</u>
- Tatem AJ, Smith DL, Gething PW, Kabaria CW, Snow RW, et al. (2010) Ranking of elimination feasibility between malaria-endemic countries. Lancet 376: 1579–1591. doi: <u>10.1016/S0140-6736(10)61301-3</u> PMID: <u>21035838</u>
- Pigott DM, Atun R, Moyes CL, Hay SI, Gething PW (2012) Funding for malaria control 2006–2010: a comprehensive global assessment. Malar J 11: 246. doi: 10.1186/1475-2875-11-246 PMID: 22839432
- Gyapong JO, Kyelem D, Kleinschmidt I, Agbo K, Ahouandogbo F, et al. (2002) The use of spatial analysis in mapping the distribution of bancroftian filariasis in four West African countries. Ann Trop Med Parasitol 96: 695–705. PMID: <u>12537631</u>
- Zoure HGM, Wanji S, Noma M, Amazigo UV, Diggle PJ, et al. (2011) The geographic distribution of *Loa loa* in Africa: results of large-scale implementation of the Rapid Assessment Procedure for Loiasis (RAPLOA). PLoS Negl Trop Dis 5: e1210. doi: <u>10.1371/journal.pntd.0001210</u> PMID: <u>21738809</u>
- Brooker S, Kabatereine NB, Gyapong JO, Stothard JR, Utzinger J (2009) Rapid mapping of schistosomiasis and other neglected tropical diseases in the context of integrated control programmes in Africa. Parasitology 136: 1707–1718. doi: 10.1017/S0031182009005940 PMID: 19450373
- Sturrock HJW, Picon D, Sabasio A, Oguttu D, Robinson E, et al. (2009) Integrated mapping of neglected tropical diseases: epidemiological findings and control implications for northern Bahr-el-Ghazal state, southern Sudan. PLoS Negl Trop Dis 3: e537. doi: <u>10.1371/journal.pntd.0000537</u> PMID: <u>19859537</u>
- Sturrock HJW, Hsiang MS, Cohen JM, Smith DL, Greenhouse B, et al. (2013) Targeting asymptomatic malaria infections: active surveillance in control and elimination. PLoS Med 10: e1001467. doi: <u>10.</u> <u>1371/journal.pmed.1001467</u> PMID: <u>23853551</u>
- 24. WHO (2014) International travel and health: 2014 updates. Geneva: World Health Organization. 248 p.
- Field VK, Ford L, Hill DR, editors (2010) Health information for overseas travel. London: National Travel Health Network and Centre. 398 p.
- CDC (2013) CDC health information for international travel 2014. New York: Oxford University Press. 688 p.
- 27. Diggle PJ, Ribeiro PJ (2010) Model-based geostatistics: Springer. 246 p.
- Hay SI, George DB, Moyes CL, Brownstein JS (2013) Big data opportunities for global infectious disease surveillance. PLoS Med 10: e1001413. doi: <u>10.1371/journal.pmed.1001413</u> PMID: <u>23565065</u>
- Gething PW, Patil AP, Smith DL, Guerra CA, Elyazar IRF, et al. (2011) A new world malaria map: *Plasmodium falciparum* endemicity in 2010. Malar J 10: 378. doi: <u>10.1186/1475-2875-10-378</u> PMID: <u>22185615</u>
- Guerra CA, Hay SI, Lucioparedes LS, Gikandi PW, Tatem AJ, et al. (2007) Assembling a global database of malaria parasite prevalence for the Malaria Atlas Project. Malar J 6: 17. PMID: <u>17306022</u>

- Hurlimann E, Schur N, Boutsika K, Stensgaard AS, Laserna de Himpsl M, et al. (2011) Toward an open-access global database for mapping, control, and surveillance of neglected tropical diseases. PLoS Negl Trop Dis 5: e1404. doi: <u>10.1371/journal.pntd.0001404</u> PMID: <u>22180793</u>
- Brooker S, Rowlands M, Haller L, Savioli L, Bundy DAP (2000) Towards an atlas of human helminth infection in sub-Saharan Africa: the use of geographical information systems (GIS). Parasitol Today 16: 303–307. PMID: <u>10858650</u>
- Hay SI, Snow RW (2006) The Malaria Atlas Project: developing global maps of malaria risk. PLoS Med 3: 2204–2208.
- Wertheim HFL, Horby P, Woodall JP (2012) Atlas of human infectious diseases, <u>https://infectionatlas.org/</u>. Oxford: Wiley-Blackwell. 280 p.
- Magill AJ, Hill DR, Solomon T, Ryan ET, editors (2013) Hunter's tropical medicine and emerging infectious disease. London: Elsevier. 1190 p.
- Farrar JJ, Hotez PJ, Junghanss T, Kang G, Lalloo D, et al., editors (2014) Manson's tropical diseases. London: Elsevier. 1337 p.
- Guernier V, Hochberg ME, Guegan JFO (2004) Ecology drives the worldwide distribution of human diseases. PLoS Biol 2: 740–746.
- Dunn RR, Davies TJ, Harris NC, Gavin MC (2010) Global drivers of human pathogen richness and prevalence. Proc R Soc Lond B Biol Sci 277: 2587–2595.
- Jones KE, Patel NG, Levy MA, Storeygard A, Balk D, et al. (2008) Global trends in emerging infectious diseases. Nature 451: 990–994. doi: <u>10.1038/nature06536</u> PMID: <u>18288193</u>
- 40. Moyes CL, Henry AJ, Golding N, Huang Z, Singh B, et al. (2014) Defining the geographical range of the Plasmodium knowlesi reservoir. PLoS Negl Trop Dis 8: e2780. doi: <u>10.1371/journal.pntd.0002780</u> PMID: <u>24676231</u>
- 41. Barber BE, William T, Grigg MJ, Menon J, Auburn S, et al. (2013) A prospective comparative study of knowlesi, falciparum, and vivax malaria in Sabah, Malaysia: high proportion with severe disease from *Plasmodium knowlesi* and *Plasmodium vivax* but no mortality with early referral and artesunate therapy. Clin Infect Dis 56: 383–397. doi: 10.1093/cid/cis902 PMID: 23087389
- Berman JJ (2012) Taxonomic guide to infectious diseases: understanding the biologic classes of pathogenic organisms. London: Elsevier. 355 p.
- 43. Brady OJ, Golding N, Pigott DM, Kraemer MU, Messina JP, et al. (2014) Global temperature constraints on Aedes aegypti and Ae. albopictus persistence and competence for dengue virus transmission. Parasit Vectors 7: 338. doi: 10.1186/1756-3305-7-338 PMID: 25052008
- 44. GBD 2013 Disease and Injury Incidence and Prevalence Collaborators (2015) Global, regional, and national incidence, prevalence and YLDs for 301 acute and chronic diseases and injuries for 188 countries, 1990–2013: a systematic analysis for the Global Burden of Disease Study 2013. Under submission.
- 45. GBD 2013 Mortality and Causes of Death Collaborators (2015) Global, regional, and national age-sex specific all-cause and cause-specific mortality for 240 causes of death, 1990–2013: a systematics analysis for the Global Burden of Disease Study 2013. Lancet 385: 117–171. doi: <u>10.1016/S0140-6736(14)</u> <u>61682-2</u> PMID: <u>25530442</u>
- 46. WHO International statistical classification of diseases and related health problems 10th revision. <u>http://apps.who.int/classifications/icd10/browse/2010/en</u>. Accessed: June 2014
- Edberg SC (2005) Global infectious diseases and epidemiology network (GIDEON): a world wide webbased program for diagnosis and informatics in infectious diseases. Clin Infect Dis 40: 123–126. PMID: <u>15614701</u>
- Hirsch JE (2005) An index to quantify an individual's scientific research output. Proc Natl Acad Sci USA 102: 16569–16572. PMID: <u>16275915</u>
- 49. Scopus Homepage. <u>www.scopus.com</u>. Accessed: July 2014
- 50. McIntyre KM, Hawkes I, Waret-Szkuta A, Morand S, Baylis M (2011) The *h*-Index as a quantitative indicator of the relative impact of human diseases. PLoS One 6: e19558. doi: <u>10.1371/journal.pone.</u> 0019558 PMID: <u>21625581</u>
- Murray CJ, Ortblad KF, Guinovart C, Lim SS, Wolock TM, et al. (2014) Global, regional, and national incidence and mortality for HIV, tuberculosis, and malaria during 1990–2013: a systematic analysis for the Global Burden of Disease Study 2013. Lancet.
- WHO (2012) Accelerating work to overcome the global impact of neglected tropical diseases—a roadmap for implementation. Geneva: World Health Organization. 16 p.

- Hotez PJ, Alvarado M, Basanez MG, Bolliger I, Bourne R, et al. (2014) The Global Burden of Disease study 2010: interpretation and implications for the Neglected Tropical Diseases. PLoS Negl Trop Dis 8: e2865. doi: 10.1371/journal.pntd.0002865 PMID: 25058013
- 54. Southwood TRE, Henderson PA (2000) Ecological methods. Oxford: Wiley-Blackwell. 592 p.
- 55. Malaria Atlas Project Homepage. http://www.map.ox.ac.uk/. Accessed: July 2014
- 56. Clements ACA, Reid HL, Kelly GC, Hay SI (2013) Further shrinking the malaria map: how can geospatial science help to achieve malaria elimination? Lancet Infect Dis 13: 709–718. doi: <u>10.1016/S1473-</u> 3099(13)70140-3 PMID: 23886334
- Hay SI, Smith DL, Snow RW (2008) Measuring malaria endemicity from intense to interrupted transmission. Lancet Infect Dis 8: 369–378. doi: 10.1016/S1473-3099(08)70069-0 PMID: 18387849
- 58. The DHS Program Homepage. http://dhsprogram.com/. Accessed: July 2014
- 59. Moyes CL, Temperley WH, Henry AJ, Burgert CR, Hay SI (2013) Providing open access data online to advance malaria research and control. Malar J 12: e161.
- The DHS Program HIV/AIDS survey indicators database. <u>http://hivdata.dhsprogram.com/</u>. Accessed: July 2014
- UNICEF Multiple indicator cluster survey (MICS). <u>http://www.unicef.org/statistics/index_24302.html</u>. Accessed: July 2014
- 62. WHO (2013) Global tuberculosis report 2013. Geneva: World Health Organization. 289 p.
- Pullan RL, Gething PW, Smith JL, Mwandawiro CS, Sturrock HJW, et al. (2011) Spatial modelling of soil-transmitted helminth infections in Kenya: a disease control planning tool. PLoS Negl Trop Dis 5: e958. doi: 10.1371/journal.pntd.0000958 PMID: 21347451
- WHO (2010) Working to overcome the global impact of neglected tropical diseases. First WHO report on neglected tropical diseases. Geneva: World Health Organization. 172 p.
- 65. GPEI (2013) Eighth report—October 2013. London: Global Polio Eradication Initiative. 59 p.
- Upfill-Brown AM, Lyons HM, Pate MA, Shuaib F, Baig S, et al. (2014) Predictive spatial risk model of poliovirus to aid prioritization and hasten eradication in Nigeria. BMC Med 12: 92. doi: <u>10.1186/1741-</u> <u>7015-12-92</u> PMID: <u>24894345</u>
- Andre M (2013) Assessing the risks for Poliovirus outbreaks in polio-free countries—Africa 2012–2013. MMWR Morb Mortal Wkly Rep 62: 768–772. PMID: 24048153
- WHO (2013) Control and surveillance of human African trypanosomiasis: report of a WHO expert committee. Geneva: World Health Organization. 237 p.
- Eberhard M (2013) Progress toward elimination of onchocerciasis in the Americas-1993-2012. MMWR Morb Mortal Wkly Rep 62: 405–408. PMID: 23698606
- Balabanova Y, Gilsdorf A, Buda S, Burger R, Eckmanns T, et al. (2011) Communicable diseases prioritized for surveillance and epidemiological research: results of a standardized prioritization procedure in Germany, 2011. PLoS One 6: e25691. doi: 10.1371/journal.pone.0025691 PMID: 21991334
- DISCONTOOLS Project (2012) Approaches to the prioritisation of diseases to focus and prioritise research in animal health: a worldwide review of existing methodologies. Brussels: IFAH-Europe. 18 p.
- Gilsdorf A, Krause G (2011) Prioritisation of infectious diseases in public health: feedback on the prioritisation methodology, 15 July 2008 to 15 January 2009. Euro Surveill 16: 15–21.
- 73. Krause G, Prioritisation Working Group (2008) How can infectious diseases be prioritized in public health? A standardized prioritization scheme for discussion. EMBO Rep 9: S22–S27. doi: <u>10.1038/embor.2008.76</u> PMID: 18578019
- WHO (2006) Setting priorities in communicable disease surveillance. Geneva: World Health Organization. 29 p.
- 75. Byass P, de Courten M, Graham WJ, Laflamme L, McCaw-Binns A, et al. (2013) Reflections on the Global Burden of Disease 2010 estimates. PLoS Med 10: e1001477. doi: <u>10.1371/journal.pmed.</u> <u>1001477</u> PMID: 23843748
- 76. de Martel C, Ferlay J, Franceschi S, Vignat J, Bray F, et al. (2012) Global burden of cancers attributable to infections in 2008: a review and synthetic analysis. Lancet Oncol 13: 607–615. doi: <u>10.1016/S1470-</u> 2045(12)70137-7 PMID: <u>22575588</u>
- 77. Bonds MH, Keenan DC, Rohani P, Sachs JD (2010) Poverty trap formed by the ecology of infectious diseases. Proc R Soc Lond B Biol Sci 277: 1185–1192.
- Hotez PJ (2013) Forgotten people, forgotten diseases: the neglected tropical diseases and their impact on global health and development. Washington DC: ASM Press. 275 p.
- Molyneux DH (2014) Neglected tropical diseases: now more than just 'other diseases'—the post-2015 agenda. Int Health 6: 172–180. doi: <u>10.1093/inthealth/ihu037</u> PMID: <u>24969646</u>

- Bannister B (2010) Viral haemorrhagic fevers imported into non-endemic countries: risk assessment and management. Brit Med Bull 95: 193–225. doi: <u>10.1093/bmb/ldq022</u> PMID: <u>20682627</u>
- 81. WHO (2013) World malaria report 2013. Geneva: World Health Organization. 255 p.
- 82. UNAIDS UNAIDS. http://www.unaids.org/en/. Accessed: July 2014
- Global Atlas of Helminth Infections GAHI: global atlas of helminth infections. <u>http://www.thiswormyworld.org/</u>. Accessed: July 2014
- 84. Simarro PP, Cecchi G, Paone M, Franco JR, Diarra A, et al. (2010) The Atlas of human African trypanosomiasis: a contribution to global mapping of neglected tropical diseases. Int J Health Geogr 9: e57.
- **85.** Cano J, Rebollo MP, Golding N, Pullan RL, Crellen T, et al. (2014) The global distribution and transmission limits of lymphatic filariasis: past and present. PLoS Negl Trop Dis under submission.
- GPEI Data and monitoring. <u>http://www.polioeradication.org/Dataandmonitoring.aspx</u>. Accessed: July 2014
- 87. African Programme for Onchocerciasis Control Country profiles. <u>http://www.who.int/apoc/countries/en/</u>. Accessed: July 2014
- Zoure HGM, Noma M, Tekle AH, Amazigo UV, Diggle PJ, et al. (2014) The geographic distribution of onchocerciasis in the 20 participating countries of the African Programme for Onchocerciasis control: (2) pre-control endemicity levels and estimated number infected. Parasit Vectors 7: 326. doi: <u>10.1186/</u> 1756-3305-7-326 PMID: 25053392
- 89. Trachoma Atlas Global atlas of trachoma. http://www.trachomaatlas.org/. Accessed: July 2014